 vc/12-Jul-05
Week 28  starting 11 July 05
SM18 Operation Meeting (Tuesdays)

Bench status  on Monday  11 July 05 @: 7/11/2005 4:45:19 PM
	3
	A1
	MBBR3365
	PT 12 Warm Up
	187.62

	8
	A2
	MBAR3306
	ICS 4 Final connection
	297.86

	11
	B1
	SSS174
	Prep 5 Cool Down
	296.7

	4
	B2
	SSS204
	Prep 3 CDW1 HV
	298.04

	2
	C1
	MBAL1245
	Prep 0 Scan Bench
	296.88

	10
	C2
	MBBL3156
	Prep 5 Cool Down
	242.66

	7
	D1
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	[image: image2.png]



	297.32

	12
	D2
	SSS641
	Finished Ready to go Final scan
	297.41

	1
	E1
	MBBR3292
	Prep 0 Scan Bench
	298.04

	5
	E2
	MBAL1244
	PT 5.1 1.5 KA Quench LF
	2.08 HV

	6
	F1
	MBAL3114
	Prep 5 Cool Down
	4.58 HV

	9
	F2
	MBAR3349
	ICS 4 Final connection
	296.43


   
 
Coordinator this week: John, Stephane +Marco;MTM support 16-3602
16 complete Magnet Tests cycles   (no ROBTC this week )
12  dipoles( 2 with MM )+ 3  SSS( 0  with MM) +  1 SSSS 
   Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· General Points  :
WEEKEND ISSUES :

· Problems on B2 leak & labelling  : done 3 times ?
· Not enough magnets or rather people to mount MRB’s from ICS

·  AOB
Tests 

	Magnets under Test from 03-07-2005 to 09-07-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBBL3095 
	[image: image3.png]



	


	02-07-2005 
	42 h.
	25 h.
	31 h.
	15 h.
	13 h.
	08-07-2005 
	126 h.
	Stripping
	
	

	2
	TBA2
	MBAL2193 
	[image: image5.png]



	


	30-06-2005 
	0 h.
	24 h.
	12 h.
	14 h.
	24 h.
	03-07-2005 
	74 h.
	Stripping
	
	

	3
	TBA2
	MBAR3372 
	[image: image7.png]



	


	03-07-2005 
	36 h.
	33 h.
	25 h.
	26 h.
	10 h.
	09-07-2005 
	130 h.
	Stripping
	
	

	4
	TBB1
	SSS148 
	[image: image9.png]



	


	30-06-2005 
	37 h.
	16 h.
	15 h.
	8 h.
	38 h.
	05-07-2005 
	114 h.
	Stripping
	
	

	5
	TBB1
	SSS282 
	[image: image11.png]



	


	05-07-2005 
	45 h.
	20 h.
	21 h.
	6 h.
	13 h.
	09-07-2005 
	105 h.
	Stripping
	
	

	6
	TBB2
	SSS205 
	[image: image13.png]



	


	02-07-2005 
	30 h.
	14 h.
	26 h.
	26 h.
	18 h.
	07-07-2005 
	114 h.
	Stripping
	
	

	7
	TBC1
	MBBR3314 
	[image: image15.png]



	


	02-07-2005 
	30 h.
	23 h.
	18 h.
	14 h.
	14 h.
	06-07-2005 
	99 h.
	Stripping
	
	

	8
	TBC2
	MBAR3302 
	[image: image17.png]



	


	29-06-2005 
	32 h.
	33 h.
	18 h.
	17 h.
	29 h.
	05-07-2005 
	129 h.
	Stripping
	
	

	9
	TBC2
	MBBR3366 
	[image: image19.png]



	


	05-07-2005 
	36 h.
	23 h.
	18 h.
	19 h.
	10 h.
	09-07-2005 
	106 h.
	Stripping
	
	

	10
	TBD2
	SSS640 
	[image: image21.png]



	


	29-06-2005 
	44 h.
	16 h.
	43 h.
	37 h.
	47 h.
	06-07-2005 
	187 h.
	Stand By
	
	

	11
	TBE1
	MBBR3273 
	[image: image23.png]



	


	01-07-2005 
	23 h.
	21 h.
	14 h.
	13 h.
	20 h.
	05-07-2005 
	91 h.
	Stripping
	
	

	12
	TBE2
	MBBL2161 
	[image: image25.png]



	


	28-06-2005 
	51 h.
	28 h.
	17 h.
	18 h.
	20 h.
	04-07-2005 
	134 h.
	Stripping
	
	

	13
	TBE2
	MBBR3359 
	[image: image27.png]



	


	04-07-2005 
	53 h.
	22 h.
	16 h.
	15 h.
	14 h.
	09-07-2005 
	120 h.
	Stripping
	
	

	14
	TBF1
	MBAL2051 
	[image: image29.png]



	


	03-07-2005 
	33 h.
	23 h.
	36 h.
	19 h.
	13 h.
	09-07-2005 
	124 h.
	Stand By
	
	

	15
	TBF1
	MBBR3308 
	[image: image31.png]



	


	29-06-2005 
	30 h.
	27 h.
	19 h.
	15 h.
	10 h.
	03-07-2005 
	101 h.
	Stripping
	
	

	16
	TBF2
	MBAR3321 
	[image: image33.png]



	


	30-06-2005 
	29 h.
	25 h.
	35 h.
	15 h.
	9 h.
	05-07-2005 
	113 h.
	Stripping
	
	


Quench Information

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBL3095
	11360 A.
(D1_U_S01)
	12569 A.
(D1_L_S01)
	12782 A.
(D2-L-S01)
	[image: image35.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAL2193
	12076.7 A.
(D1_L)
	12150 A.
(D2_U)
	[image: image36.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBAR3372
	9464.7 A.
(D1-U)
	11606.1 A.
(D1-U)
	11870 A.
(D1-L)
	11986 A.
(D2-L)
	12536 A.
(D2-U)
	12850 A.
[image: image37.jpg]



	-
	-
	-
	-
	-
	-
	-

	4
	SSS148
	11803.9 A.
(QE_12)
	12402.7 A.
(QE-12)
	12583.7 A.
(QE_12)
	[image: image38.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS282
	12020 A.
(QE-12)
	12579 A.
(QE-34)
	12747 A.
(QE-12)
	[image: image39.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS205
	11749 A.
(QE_34)
	12409 A.
(QE_12)
	12574 A.
(QI_34)
	[image: image40.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	MBBR3314
	11287 A.
(D1_U)
	12365 A.
(D2_L)
	12850 A.
[image: image41.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBAR3302
	12288.1 A.
(D1_U)
	12504.7 A.
(D2_U)
	12750.6 A.
(D1-L)
	[image: image42.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBR3366
	11758 A.
(D2_U)
	12650 A.
(D1-U)
	12850 A.
[image: image43.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	SSS640
	4807 A.
(MQ_Ex_34)
	5377 A.
(MQ_In_34)
	5817 A.
(MQ_In_34)
	5820 A.
[image: image44.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBBR3273
	11881.8 A.
(D2_L)
	12859 A.
[image: image45.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBBL2161
	11115.9 A.
(D2-L)
	11781.6 A.
(D2_U)
	12392 A.
(D1-L)
	12507.5 A.
(D1_U)
	[image: image46.png]



	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBBR3359
	11193.1 A.
(D1_L)
	11077 A.
(D1-U)
	12860 A.
(D1-U)
	[image: image47.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBAL2051
	10888 A.
(D2-L)
	11952 A.
(D1-L)
	12223 A.
(D1-U)
	12364 A.
(D1-U)
	12653 A.
(D1-U)
	12659 A.
(D1-U)
	12664 A.
(D1-U)
	12669 A.
(D1-U)
	12671 A.
(D1_U)
	[image: image48.png]



	-
	-
	-

	15
	MBBR3308
	1177 A.
(D1-L)
	12236.5 A.
(D1-L)
	12817 A.
(D2_U)
	[image: image49.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	16
	MBAR3321
	12125 A.
(D2-U-S01)
	12525 A.
(D1-L-S01)
	12738 A.
(D2-L-S01)
	[image: image50.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
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	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _05_N
	TBD2


	after wire installation problem with motor control in y-axis for table A
	MTM
	
	Guy checked and found that the config-files were lost. He re-launched the scripts, made some tests and then OK.

	Tuesday _05_N
	TBE1


	One of the voltage tap wires has broken too far inside the CFB to be repaired. ICS will carry on connecting.
	MTM
	
	Proceed as is.

	Tuesday _05_N
	TBD1


	ACR team indicated an alarm on the temperature of the anti-cryostat short in aperture 2 (4.7 K). The history plot shows that it gradually decreased in temperature
	MTM
	
	Investigation by F.Flamand and J.L Ducret. Nothing strange found. Measurements the day after . Temperature became normal (17 C).

	Wednesday _06_M
	TBA1


	MM2 could not be launched from TEMA. 
	MTM
	
	Giorgio and Maryline investigated. Config file not correct in TEMA.

	Wednesday _06_M
	TBF


	Variac for adjusting the Voltage for minimum energy found to work badly (rotation 1/4 turn only).
	MTM
	
	Guy informed. Check on progress.

	Wednesday _06_A
	TBB1


	SSS282:  ICS was asked to check the corrector circuit connection.
	ICS
	
	?

	Thursday _07_M
	TBD1


	4th chassis of potaim cards to be changed. 
	AB/CO
	
	Guy contacted.

	Thursday_07_A
	TBF1


	MB2051: Quench at 4.4 K. Regulation problems for the two quenches. Difficulty to have the 4.4 K cryo OK. Process problem or stability problem? Results not conform to expectation (9000 A for Iquench)
	ACR
	
	Process not automatised anymore. 

To check in another magnet.

	Thursday _07_A
	TBA2


	Communication problem between server and PLC
	AB/CO?
	
	

	Thursday _07_N

Friday _08_A
	TBB2


	LF acquisition was not getting armed as some triggers existing, while launching prep 2.2c 
Again some triggers


	ICS
	
	ICS re-opened the M3 line and checked the connection of the EF clamp. A wire was found broken.

SSS with new configuration of connection. But not in the list given to OP team.

ICS had managed to connect it using the old schema??

	Thursday _07_N
	TBB2


	HF window is not getting displayed  in the local server during the IAP at cold. 
	AB/CO
	hf
	Rebooting the sun test could be carried out in normal conditions

	Thursday _07_N
	TBC1


	Lf 1.5 kA quench: flat signal for D1-D2.

Cable for this potaim card disconnected. Not reconnected after the offsets adjustments of the potaim cards
	MTM
	
	Cable re-connected and test program  went on.

Lf quench checked by Stephane Friday. OK given for warming up.

	Saturday _09_A
	TBC1


	MBB2197: After each quench (training 2) the vacuum degrades (10-4 mbar) not seen at warm.
	
	
	Test stop: investigation by ACR team if it come from magnet or CFB.

	Saturday _09_A
	TBD2


	Disconnection of the SSS641 by ICS with rotary switch put in D2!
	MTM
	
	Not allowed for safety reasons. Switch put in CC.

	Saturday _09_A
	TBA2


	MB3349 (with MM) put on TBA2 and not on TBF2 as foreseen.
	ICS
	
	Rocla and ICS informed. TC also contacted.

	Saturday_09_N
	TBC2

TBB2


	Scan cannot be performed with the MOBIC (MB3156). 

Same problem for SSS on TBB2
	MTM
	
	Magnet name entered manually.

Guy came and found that the battery had a problem: action on underway.

	Sunday_10_M
	TBB2


	Leak on M2 line of SSS204 
	ICS
	
	Joint changed but leak remained. M2 line and M3 line inverted. Leak in M3 line! Magnet to be re-connected.

	Sunday_10_A
	TBA1


	During MMP2 LHC cycle, the rotation goes to the end stop for aperture 2 
	MTM
	
	The TRU cover is touching the flat cables. Guy advised to proceed.

	Sunday_10_N
	TBF2


	No communication between PLCSE and server.
	AB/CO
	
	Exit from the server reset the PLCSE.

	Monday_11_M
	TBA


	Hf DAQ could not be armed.
	AB/CO
	
	

	Monday_11_M
	TBE2


	Trigger in Potaim card D2_L because of a significant offset.
	MTM
	
	Maryline informed.

 Maryline changed the card.











